Author's response to reviews

Title: Duplication involving GABA receptor genes accounts for paradoxical response to antiepileptic drugs in Isodicentric 15 syndrome.

Authors:

Alessandro Di Rocco (ALESSANDRO.DIROCCO@NYUMC.ORG)
Andrea Loggini (andrealoggini@gmail.com)
Maja Di Rocco (majadirocco@ospedale-gaslini.ge.it)
Pietro Di Rocco (pdl.rocco@libero.it)
Roger Rossi (RRossi@JFKHealth.org)
Giorgio Gimelli (giorgio.gimelli@gmail.com)
Carl Bazil (cwb11@mail.cumc.columbia.edu)

Version: 2 Date: 22 November 2012

Author's response to reviews:

Dear Dr. Majithia,

please consider our manuscript for publication in BMC Neurology.

We believe that our report for the first time correlates specific abnormalities of genes encoding for GABA receptors with an abnormal or paradoxical response to GABAergic drugs in an adult with isodicentric 15 syndrome (IDIC-15).

To our knowledge this is also the first report of any treatment or treatment complication ever reported in an individual affected by IDIC-15.

We believe that even an individual case report may be extremely valuable for clinicians treating children and adults with this rare genetic disorder and we are aware that the IDIC-15 community is eager to learn of any experience that could help define a better treatment of these individuals.

We hope that BMC Neurology will kindly consider this report for publication.

We added the "Patient informed consent" as requested.

My kindest regards,

Alessandro Di Rocco, MD
Professor of Neurology & Chief, Division of Movement Disorders
NYU Langone Medical Center - Parkinson and Movement Disorders Center
New York University School of Medicine
145 E 32nd Street